Author's response to reviews

Title: Using automated medical records for rapid identification of illness syndromes: the example of lower respiratory infection

Authors:

- Ross Lazarus (ross.lazarus@channing.harvard.edu)
- Kenneth P Kleinman (ken_kleinman@harvardpilgrim.org)
- Inna Dashevsky (inna_dashevsky@harvardpilgrim.org)
- Alfred DeMaria Jr. (alfred.demaria@state.ma.us)
- Richard Platt (richard.platt@channing.harvard.edu)

Version: 4 Date: 13 Aug 2001

As requested, figures 3, 4, and 5 have been revised to improve legibility.